Title: Frequency and impact of selection bias in observational studies assessing influenza vaccine effectiveness: A systematic review

Authors:

Cornelius Remschmidt (remschmidtc@rki.de)
Ole Wichmann (wichmanno@rki.de)
Thomas Harder (hardert@rki.de)

Version: 2 Date: 18 March 2015

Author's response to reviews:

Dear Editor,

as requested, we made the following changes to the manuscript:
1) The email addresses of all authors were included on the title page.
2) The Introduction was renamed as Background.
3) An Acknowledgment section was included.
4) Tables and figure legends were moved after the Reference section.
5) Figure 3A and B was divided into Figure 3 and 4 and all respective references in the text were changed.

We hope that in this form the manuscript is ready for peer review.

With kind regards

Thomas Harder (on behalf of all co-authors).